Author's response to reviews

Title: Pulmonary emangioendothelioma presenting with vertebral metastases: a case report

Authors:

- Angela Sardaro (angela.sardaro@uniba.it)
- Lilia Bardoscia (liliabardoscia@gmail.com)
- Maria Fonte Petruzzielli (mari.petruzzielli@gmail.com)
- Anna Nikolaou (annanikolaou@hotmail.it)
- Beatrice Detti (beatrice.detti@unifi.it)
- Giuseppe Angelelli (giuseppe.angelelli@uniba.it)

Version: 2  Date: 21 February 2014

Author's response to reviews:

We revised the presentation of our report in order to give you more details about our purpose: firstly, we believe this report is the first one of this kind in the literature: several papers about epithelioid hemangioendothelioma are available, none of them focusing on the role of Radiation Therapy in so detailed and precise way as we do. Secondly, we believe that this report might have a broader clinical impact across more than one area of medicine: the available literature shows that, because of its heterogeneous presentation, epithelioid hemangioendothelioma may get pneumologist attention or the orthopaedic one, vascular surgeon, thoracic surgeon, internist, even dermatologist attention.